EE 150 Quiz #0

Due: Oct. 11, 1996 2:30 pm (in class)

There will be a series of in-class quizes given out, the times will not be announced. Some will be done in-class, while others will be due over a 1 week period. These are in addition to ongoing assignments.

This is a sample, and for this week only it is anonymous (don’t write your name on this sheet anywhere!). Please answer the following questions as best you can by marking the appropriate selection in the range shown. Also, add comments as you wish.

The length of the midterm LONG 1 2 3-fine 4 5 SHORT Comment:

The difficulty of the midterm HARD 1 2 3-fine 4 5 EASY Comment:

The length of the assignments LONG 1 2 3-fine 4 5 SHORT Comment:

The difficulty of the assignments HARD 1 2 3-fine 4 5 EASY Comment:

The class is moving too QUICKLY 1 2 3-fine 4 5 SLOWLY Comment:

Examples in class are POOR/USELESS 1 2 3-fine 4 5 EXCELLENT Comment:

Compared to other class, this class is HARD 1 2 3-fine 4 5 EASY Comment:

Out-of-class work vs other classes WAY-MORE 1 2 3-same 4 5 WAY-LESS Comment:

The text is POOR/USELESS 1 2 3-fine 4 5 VERY-USEFUL Comment:

Table 1: Check off how you interpret each of the above.

Please add any other specific comments here or on the back regarding the midterm or the course.

For midterm 1, I worked through both sample exams TRUE FALSE Comment:

For mini-assignment 3, I through the problem mostly alone TRUE FALSE Comment:

I would be capable of doing mini-assig 3 on my own TRUE FALSE Comment:

I generally have read the material before the lecture TRUE FALSE Comment:

I never read the material before the lecture TRUE FALSE Comment:

I took EE150 because I want to learn C TRUE FALSE Comment:

I took EE150 because I have to take the class TRUE FALSE Comment:

Table 2: Check off true/false for your opinion on each of the above.

Table 3: Write down any additional comments.